Author’s response to reviews

Title: Health facility committees and facility management - exploring the nature and depth of their roles in Coast Province, Kenya

Authors:

Catherine Goodman (catherine.goodman@lshtm.ac.uk)
Antony Opwora (opwora@gmail.com)
Margaret Kabare (maggikabare@yahoo.com)
Sassy Molyneux (smolyneux@kilifi.kemri-wellcome.org)

Version: 4 Date: 1 September 2011

Author’s response to reviews:

Dear Editors

Thanks for your email of 31 August 2011, requesting some additional formatting changes to the tables in our paper.

We have:
- converted table one from a box to a table format
- on Table 4 we have split any cells which were merged across rows.

We now believe that all our tables meet your requirements, but please let us know if any further adaptations are needed.

yours faithfully

Catherine Goodman